Author's response to reviews

Title: Disagreement in primary study selection between systematic reviews on negative pressure wound therapy

Authors:

Frank Peinemann (frank.peinemann@iqwig.de)
Natalie McGauran (n.mcgauran@iqwig.de)
Stefan Sauerland (stefan.sauerland@ifom-uni-wh.de)
Stefan Lange (stefan.lange@iqwig.de)

Version: 6 Date: 25 June 2008

Author's response to reviews:

Formatting changes:

Text removed from the foot of each page: 'Revision-2. Subject: 1416708205176963 - Disagreement in primary study selection'

'Figures' re-titled 'Figure legends'.

Figures cropped.